Author's response to reviews

Title: Neonatal outcomes in obese mothers: A population-based analysis

Authors:
Anne-Frederique Minsart (aminsart@ulb.ac.be)
Pierre Buekens (pbuekens@tulane.edu)
Myriam De Spiegelaere (mdespiegelaere@ggc.irisnet.be)
Yvon Englert (yvon.englert@erasme.ulb.ac.be)

Version: 2 Date: 8 October 2012

Author's response to reviews:

Dear Editor,
"We have added the following paragraph in the Methods section as suggested:
The register is anonymous and publicly available, and accessible with the permission of the Brussels Health Observatory and the Health Department of the French Community of Belgium."
Dr Anne-Frederique Minsart